Author's response to reviews

Title: General practitioners' conceptions of depression and professional competence. A postal survey.

Authors:

Stig J Andersson (gits.joel@tele2.se)
Margareta Troein (margareta.troein@smi.mas.lu.se)
Gunnar Lindberg (gunnar.lindberg@nepi.net)

Version: 2 Date: 13 January 2005

Author's response to reviews:

We want to add two acknowledgements to the article:
1. The County council of Varmland for financial support.
2. Professor Staffan Jansson and his staff at the research department of Primary Care, Karlstad.